Author’s response to reviews

Title: Virulence characteristics of multidrug resistant biofilm forming Acinetobacter baumannii isolated from intensive care unit patients

Authors:
Habib Zeighami (zeighami@zums.ac.ir)
Fatemeh Valadkhani (f.valadkhani@yahoo.com)
reza Shapouri (rezashapoury@yahoo.com)
Elham Samadi (elhamsamadi.microbiology@gmail.com)
Fakhri Haghi (haghi@zums.ac.ir)

Version: 2 Date: 09 Jul 2019

Author’s response to reviews:

Dear Oliver Cocks

BMC Infectious Diseases

We edited our manuscript point by point according to editor comments.

Technical Comments:

Editor Comments:

1. Please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your "Ethics and Consent to Participate" section. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an ethics committee or is deemed unnecessary according to national regulations, please clearly state this, including the name of the committee or a reference to the relevant legislation.

Response: this section edited as:

Ethical clearance was obtained from the Ethics Committee of Zanjan University of Medical Sciences before conducting the study (IR.ZUMS.REC.1396.278). All participants provided written informed consent to participate in the studies.
2. In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data.
Response: this section edited as:
The datasets will not be available on a publically available website, but it may be possible to provide access to anonymized data. Anyone who wants to request the data can contact with Fakhri Haghi, corresponding author.

3. Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.
Response: this section edited according to comment.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response: We upload our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

Reviewer reports:
Juan J Camarena, PhD, MD (Reviewer 2): The modifications and clarifications made by the authors before the issues raised in the first revision, are the correct ones. Thank you